Author's response to reviews

Title: Rationale and design of the PREDICE project: Cost-Effectiveness of Type 2 diabetes prevention among high-risk Spanish individuals following lifestyle intervention in real-life primary care setting.

Authors:

Bernardo Costa (costaber@gmail.com)
Juan J Cabre (jcabre.tarte.ics@gencat.cat)
Ramon Sagarra (rsagarralamo@gmail.com)
Oriol Sola-Morales (osola@sabirmedical.com)
Francesc Barrio (ciscobarrio@gmail.com)
Josep L Piñol (joseplluispinol@gmail.com)
Xavier Cos (xcos.claramunt@gmail.com)
Bonaventura Bolibar (bbolibar@idiapigol.org)
Conxa Castell (conxa.castell@gencat.cat)
Katarzyna Kissimova-Skarbek (katia.skarbek@idf.org)
Jaakko Tuomilehto (jaakko.tuomiletho@ktl.fi)
Research Group DE-PLAN-CAT/PREDICE (bcosta.preus@scs.es)

Version: 3 Date: 28 July 2011

Author's response to reviews: see over
Dear Editorial Production Team,

We revised the manuscript “Rationale and design of the PREDICE project: Cost-Effectiveness of Type 2 diabetes prevention among high-risk Spanish individuals following lifestyle intervention in real-life primary care setting” (MS: 6374513085697523) and we enclose herewith the changes. Please note that your remarks have been carefully taken into account.

Regarding your detailed remarks:

**Major revisions**

**Author list:** the list of authors in the manuscript should be written exactly as they are in the submission system, both in style and order. The preferred style is 'First name Initial Last name' (e.g. Joe F Bloggs) **Done**

**Affiliations:** Please number all 5 of your affiliations and ensure there is a numeric link to every author in the author list **Done**

**Structure:** this manuscript does not have the correct structure. Please reformat it to use these headings: Abstract; Background; Methods/Design; Discussion; Abbreviations (if any); Competing interests; Authors' contributions; Authors' information (optional); Acknowledgements (optional); References

The article has been formatted according to the proposed guidelines.

**References:** Please make sure references are cited in numerical order. We notice references 10 and 11 are not cited within the main text. Please make sure references continue to be cited in numerical order once this reference is cited. **Done**

**Figures:** All figures must be cited; figures not cited will be excluded from the final manuscript. Please ensure they are cited in numerical order **Done**

**Figures:** Please provide figure titles. These should be concise and be the first sentence of the figure legend these should be placed in a figure legends section section after the references **Done**

**Minor revisions**

**Highlighting/tracking:** Please remove all highlighting and tracking from the manuscript. **Done**

**Title page:** Please remove "(*) A complete list of members of the coordinating committee and participating centres is given at the end of the article." from the title page **Done**

Please place the "Members of the DE-PLAN-Catalonia Coordinating Committee" and "Participating reference centres:" in an acknowledgements section **Done**

**References:** the reference list should contain up to 30 authors' names for each citation. The term 'et al.' should not be used if there are fewer than 30 authors. **Done**

**Figures:** The image file should not include the title (e.g. Figure 1... etc.) or figure number. **Done**
Figures: It is important for the final layout of the manuscript that the figures are cropped as closely as possible to minimise white space around the image. Our online figure guide contains full details for preparing files for submission and can be viewed here: http://www.biomedcentral.com/info/ifora/figures

Additional files: Please remove additional files

Typography: Please take this opportunity to check your manuscript for any typographical errors and to make any final corrections or revisions. This is the final proofing stage for your manuscript, and you will not be able to make any changes after acceptance.

Please if this new version is finally found suitable for publication we would very much appreciate a letter of acceptance.

We look forward to hearing from you soon.

Dr. Bernardo Costa
Diabetes and Metabolism Research Area
Jordi Gol Primary Health Care Research Institute
Catalan Health Institute

Barcelona, 28 July 2011